Author’s response to reviews

Title: Shearing-Force Injury of a Kidney Transplant Graft During Cesarean Section: A Case Report and Review of the Literature

Authors:

Catherine Gordon (cgordon1@uci.edu)
Vasiliki Tatsis (vtatsis@uci.edu)

Version: 4 Date: 02 Mar 2019

Author’s response to reviews:

Response to Reviewers

March 2nd, 2019

Dear Editorial Board,

Thank you for the thoughtful comments on our manuscript and the opportunity to re-submit for publication in BMC Nephrology.

Regarding Figure 2, this is an original image from the authors cited. We have reached out to Elsevier and Rights Link regarding permissions for publication. The article is under a Creative Commons license. According to the license website, we are free to share the material (copy and redistribute) in any medium or format under the following terms:

Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use.

Non Commercial — You may not use the material for commercial purposes.
No Derivatives — If you remix, transform, or build upon the material, you may not distribute the modified material.

No additional restrictions — You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

Given we are using the figure solely for publication and providing appropriate credit, we believe we are within these stipulations. Here is the website for Creative Commons: https://creativecommons.org/licenses/by-nc-nd/4.0/

Sincerely,

Catherine Gordon

Vasiliki Tatsis